Author's response to reviews

Title: Prevalence and predictors of hypertension among adults in Kabul, Afghanistan

Authors:  
Islam KM Saeed (km_islam2001@yahoo.com)  
Hafez M Rasooly (dochafez@yahoo.com)

Version: 4  
Date: 27 February 2014

Author's response to reviews: see over
Cover Letter for revised version

Dear Editor in Chief,

Please find enclosed the latest revised version of our manuscript “Prevalence and predictors for hypertension among adults in Kabul, Afghanistan” by Khwaja Mir Islam Saeed et al which we would like to submit for publication as an original research article in Biomed Central Public Health. To our knowledge this is the first report showing the prevalence and associated risk factors for hypertension in Kabul the capital of Afghanistan. We believe our findings would appeals to the leadership of Biomed Central Public Health.

After series of submission and obtaining comments and feedback from the journal now we have incorporated all the comments and it has been reviewed by copy editor for English language. In addition it has been analyzed for technical issues and positive changes have been brought to this version of manuscript. For example the title is changed, title page is added and some tables have been merged. Now instead of five we have just three tables. Lots of unnecessary and feeling addition text has been removed.

Hereby we would like to confirm that our manuscript has not been published elsewhere and is not under consideration by another journal. All authors have approved the revised manuscript and agree with its submission to Biomed central public health.

As mentioned in our previous cover letter please addresses all correspondence to km_islam2001@yahoo.com or kmislamsaeed@gmail.com and look forward to hearing from you at your earliest convenience.

Sincerely,

Khwaja Mir Islam Saeed, MD, MSc

Principle Investigator and Corresponding Author